Author’s response to reviews

Title: Drug use disorder following early life exposure to tetrachloroethylene (PCE)-contaminated drinking water: a retrospective cohort study

Authors:

Ann Aschengrau (aaschen@bu.edu)
Alexandra Grippo (argrippo@buffalo.edu)
Michael R. Winter (mwinter@bu.edu)
Margaret G. Shea (mgshea@bu.edu)
Roberta F. White (rwhite@bu.edu)
Richard Saitz (rsaitz@bu.edu)

Version: 2 Date: 21 Jul 2020

Author’s response to reviews:

Response to Reviewer 2
ENHE-D-20-00193 Second Revision
“Drug use disorder following early life exposure to tetrachloroethylene-contaminated drinking water” by Ann Aschengrau, Alexandra Grippo, Michael R. Winter, Margaret G. Shea, Roberta F. White and Richard Saitz

Thank you again for your additional comments. They have been addressed as described below.

1. The p value for interaction has been added to the results section of the text.
2. A paragraph about the biological plausibility of the greater effect that was observed among females has been added to the discussion section of the text.